\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}



\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]





%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/0000-0002-5549-025X}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{???.??} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{??X??} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}



\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
     {\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 20, No. 2, pp. 144--145 (2023) \hspace{\fill}{\rm\small УДК 519.833.5}}%\udcs \hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
     	{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\ \ 91A06; 91A10} %\mscs
  }%\vbox
}



\begin{document}
\begin{thebibliography}{99} % Ссылки на литературу приводятся только на английском языке (даже если у источника нет англоязычной версии).

\bibitem[1]{aum}{R.\,J. Aumann},  {\em The core of a cooperative
game without side payments}, Transactions of the American Mathematical Society, {\bf 98}:3  (1961), 539--552.

\bibitem[2]{B-S}{A. Basile, V. Scalzo,} \href{https://doi.org/10.1007/s00182-020-00736-y}
{\em Non-emptiness of the alpha-core: sufficient and necessary conditions}, International Journal of Game Theory, {\bf 49} (2020), 1143--53.

\bibitem[3]{bon0}{O.\,N. Bondareva,} {\em The core of an $N$
person game}, Vestnik Leningrad University, {\bf 17}:13 (1962), 141--142.

\bibitem[4]{bon}{O.\,N. Bondareva,} {\em  Some applications of
linear programming to the theory of cooperative games}, Selected Russian Papers in Game Theory 1959--1965, {Princeton University Press}, Princeton, 1968, 79--114.

\bibitem[5]{FLOO}{M. Florenzano,} {\em On the
non-emptiness of the core of a coalitional production economy without ordered preferences},
Journal of Mathematical Analysis and Applications, {\bf 141} (1989), 484--490.

\bibitem[6]{Kajii}{A. Kajii,} {\em A generalization of Scarf’s theorem: An
$\alpha$-core existence theorem without transitivity or completeness}, Journal of Economic Theory, {\bf 56}:1, \mbox{194--205}.

\bibitem[7]{dan} {V.\,I. Danilov,}  {\em Lectures on Game Theory},
New Economic School, Moscow, 2002, 140~p.

\bibitem[8]{M-Y} {V.\,F. Martins-da-Rocha, N.
Yannelis,} {\em Non-emptiness of the alpha-core}, see https://www.researchgate.net/publication/241758318

\bibitem[9]{Moul}{H. Moulin,}   {\em Axioms of Cooperative
Decision Making}, Cambridge University Press, Cambridge/New\,York/New\,Rochelle/Melbourne/Sydney (1988).

\bibitem[10]{Scarf0}{H.\,E. Scarf,} {\em  The core of an $n$-person
game},{ Econometrica} {\bf 35} (1967), 50--69.

\bibitem[11]{Scarf}{H.\,E. Scarf,} {\em On the existence of a
coopertive solution for a general class of $n$-person games}, { Journal of Economic Theory}, {\bf 3}:2 (1971), 169--181.

\bibitem[12]{Shapley}{L. Shapley,} {\em On balanced sets and
cores}, RM-4601-PR, { Rand Corporation}, Santa Monica, California (1965).

\bibitem[13]{Yang}{Z. Yang,}  {\em Some infinite-player generalizations of Scarf’s
theorem: Finite-coalition $\alpha$-cores and
weak $\alpha$-cores}, {Journal of Mathematical Economics}, {\bf 73} (2017), 81--85.

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}
\newtheorem{example}{Example}%[section]
\newcommand{\beg}{\begin{equation}}
\newcommand{\eg}{\end{equation}}
\newcommand{\qq}{\qquad}
\newcommand{\q}{\quad}
\def\I{\mathcal{I}}
\newcommand{\bc}{\begin{center}}
\newcommand{\ec}{\end{center}}
\newcommand{\bs}{\setminus}
\newcommand{\Ra}{\Rightarrow}
\newcommand{\rta}{\rightarrow}
\newcommand{\la}{\langle}
\newcommand{\ra}{\rangle}
\newcommand{\ov}{\overline}
\newcommand{\empt}{\emptyset}
\newcommand{\emp}{\emptyset}
\newcommand{\sue}{\subseteq}
\newcommand{\conv}{\mathop{\rm co}\nolimits}%выпуклая оболочка
\newcommand{\R}{\mathbb{R}}
\newcommand{\ga}{\gamma}
\newcommand{\al}{\forall}





\thispagestyle{empty}

\title[On $\alpha$-core existence]{\large A Brief Proof of the Existence of an $\alpha$-Core for Generalized Strategy Games}


\author[V.\,M. MARAKULIN]{{\bf V.\,M. MARAKULIN}\protect\orcidicon{0000-0001-8827-2630}}



\address{Valeriy Mikhailovich MARAKULIN, Sobolev Institute of Mathematics,
Russian Academy of Sciences, 4 Acad. Koptyug avenue, Novo\-si\-birsk, Russia, 630090, e-mail:
marakulv@gmail.com, fax: +7(383)333 25 98, phone: +7(383)329 75 27 % Контактные данные всех авторов и место работы указываются только на английском языке
\newline\hphantom{iii} Sobolev Institute of Mathematics,
\newline\hphantom{iii} pr. Koptyuga, 4,
\newline\hphantom{iii} 630090, Novosibirsk, Russia}%
\email{\textcolor{blue}{marakulv@gmail.com}}%

\thanks{\sc MARAKULIN, V.M.,
A Brief Proof of the Existence of an $\alpha$-Core for Generalized Strategy Games}
\thanks{\copyright \ 2025 MARAKULIN V.M.}
\thanks{\rm The study was supported by the Program of Basic Scientific Research of
the Siberian Branch of the Russian Academy of Sciences (Grant no.    FWNF-2022-0019)}
\thanks{\it Received June, 3, 2025, Published December, 31, 2025}%


\semrtop \vspace{1cm}
\maketitle {\small

\vspace{-12pt}
\centerline{{\it Communicated by} {}%\sc MOTOR-2025}
%({\it filled by Editorial board})
}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:} This paper studies the problem of whether an $\alpha$-core exists (is non-empty) for generalized normal-form games where player payoffs are described by point-to-set convex-valued, ir\-ref\-le\-xi\-ve, and weakly continuous preferences. We propose a short effective proof of the non-emptiness of the core; it based on a contradiction with Brouwer's theorem on the existence of a fixed point for a continuous mapping of a convex compact set into itself.
\medskip

\noindent{\bf Keywords:} generalized strategic game, $\alpha$-core, Brouwer's fixed point theorem.
 \end{quote}
}

\bigskip

\section{Introduction}
In game theory and its applications, the concept of core is one of the key notions of the ``solution''  for any type of game model: in cooperative games specified via a characteristic function and also in a strategic (normal) form. The core includes such admissible outcomes (profile of individual strategies) that are not blocked by (admissible) coali\-ti\-ons of players---taking into account possible opposition from the members of a complementary coalition. Initially, the study of the core concept focused on games presented in characteristic function form---a mapping assigning winning options to (admissible) coali\-ti\-ons: summed payoffs (transferable utility or games with side payments) or indi\-vi\-du\-ali\-zed (non-transferable) payoffs. In \cite{bon0,bon}, Bondareva introduced balancedness, a crucial condition for the existence of core payoffs, and \cite{Shapley} later discussed it. Further,  this notion fruitfully was extended for NTU-games in  \cite{Scarf0}; one can see also \cite{Moul}, \cite{dan}.

In \cite{aum} there was first proposed the concepts of the $\alpha$-core and $\beta$-core for strategic games, with the former subsequently receiving extensive analysis in the literature. Essentially, if a coalition attempts to alter the game's current social state, a complementary coalition can counter this by adopting a strategy that harms at least one member of the original coalition, thereby preventing the change because that member would lose, and this is a way for discouraging the change. Thus, the counter-coalition destroys the cooperation of the opponents and counters the threat of changing the outcome of the game. The stability of outcomes from the $\beta$-core is much stronger---when repelling an attempt to change the outcome, the complementary coalition must have a counter-strategy that destroys the aggressor's cooperation, com\-mon to all kinds of change options. Accordingly, the $\beta$-core is a subset of the $\alpha$-core and has (much) less chance of existence.

The existence of $\alpha$-core solutions for $n$-person normal-form games was proved by \cite{Scarf} for games with quasi-concave payoff functions and a convex compact set of possible outcomes (strategy profiles). In \cite{Kajii} Scarf's existence theorem was  generalized to encompass $n$-person normal-form games with general preferences. Moreover, the description of the strategy sets in these games was as convex compact subsets of a potentially infinite-dimensional {\em normed}\/  vector space.

And this is the main drawback of Kajii's work---unlike finite-dimensional spaces, where one characterizes a compact set as norm-bounded and closed,  infinite-di\-men\-sio\-nal normed spaces lack such a simple characterization\footnote{For example, one can recall the Arzela-Ascoli theorem, which characterizes compact sets in spaces of type $C(X)$, where $X\subset \R$ is a compact subset.}.
How\-ev\-er, Kajii's proof relies heavily on compactness, otherwise the proof method used is ineffective. One can implement the latter using weak topologies and Alaoglu's theorem, where, in exactly the same way as in a finite-dimensional space, closed and bounded sets in the original normed space are weakly star-compact sets.

Paper \cite{M-Y} overcames this shortcoming---the normed strategy space---and a number of other weaknesses by using Bewley's method. This involved first proving a finite-dimensional result and then performing the limit passage over a network of finite-dimensional subspaces to transition to an infinite-dimensional Hausdorff space.
Furthermore, \cite{M-Y} clarifies the concept of the $\alpha$-core itself. In the context of a generalized game, they weaken the requirements for coalition dominance by limiting the ability of a complementary coalition to influence blocking outcomes. The existence of an element of the $\alpha$-core itself in the finite-dimensional setting is proved using the construction pro\-po\-sed
in \cite{FLOO} in the context of the Arrow-Debreu model of the economy, supplemented by one essential element. In general, the question reduces to Kakutani's theorem on the existence of a fixed point for the constructed (a cumbersome and nontrivial) point-to-set mapping. In literature there are other studies of $\alpha$-core in close to our context, for example \cite{B-S} (the study of discontinuous games), also see \cite{Yang} for infinite-player generalizations of Scarf’s theorem and so on.

The presented paper follows the setup and assumptions described in \cite{M-Y}, within which we give a new original proof of the non-emptiness of the $\alpha$-core, short and clearly demonstrating the significance of the assumptions made. Our idea of the proof is to get a contradiction with Brouwer's theorem on the existence of a fixed point for a continuous mapping of a convex compact set into itself. In the next section we describe mathematical settings, after that we formulate and prove the main result---the existence of $\alpha$-core. Presentation of an example illustrating our method of proof and conclusion remarks finish the paper.

\section{Generalized Strategic Games \label{}}
We consider a cooperative game environment in the context of a generalized normal-form game with a finite set of players (agents) ${I}=\{1,2,\ldots,n\}$. Here, any nonempty subset $S\subseteq I$ is called a coalition of players; further, we assume that in the game there are defined a set of admissible coalitions $\mathbb{I}$, such that $I$ and each single-element coalition $\{i\}$ belong to $\mathbb{I}$. Agent $i\in I$ chooses an individual strategy $x_i\in X_i$ within its own set of admissible strategies $X_i$.
\\

{\em {\bf Assumption A1.} For each agent $i\in I$, the set of strategies $X_i$ is a non\-em\-p\-ty and convex subset of the finite-dimensional topological vector space~$L$.}
\\

We call an element  $x = (x_i)_{i\in I}\in X = \prod_{i\in I} X_i$ a joint strategy (or strategic profile), and we consider it as a social state. If $S\subseteq I$, then
$X^S$ denotes the Cartesian product $\prod_{i\in S} X_i$. For simplicity, we denote the set $I\bs S=-S$ in the standard way. We denote elements of joint strategies of coalition $S \subseteq I$ as $y^S \in X^S$.
\\

{\em {\bf Assumption A2.} For each feasible coalition $S\in \mathbb{I}$, a nonempty compact set $F^S\subseteq X^S$ of feasible joint strategies exists; furthermore, the set $F=F^I$ is convex.}
\\
\def\P{\mathcal{P}}

We consider the case where agents have (possibly unordered) preferences, in the presence of externalities (i.e., preference interdependence). Formally, each player $i$ has a preference relation on $X$, which is described by a cor\-res\-pon\-den\-ce $\P_i$ from $X$ to $X$. For any outcome $x\in X$, the set $\P_i(x)$ is the set of allocations $y\in X$ that are strictly preferred by player $i$ to profile $x$.
We make the standard assumption of convex irreflexivity of preferences.
\\

{\em {\bf Assumption A3.} For each $x\in X$ and $i\in I$, $x \notin \conv \P_i(x)$ holds.}
\\

Thus, the generalized game under study is represented by the following set of parameters
$$
G =\la L, (X_i, \P_i)_{i\in
I}, F, (F^S)_{S\in \mathbb{I}}\ra.
$$

Let $F_i$ be the set of all individual strategies $z_i\in X_i$ of player $i$
that can be applied in $z = (z_j)\in F^T$, where
$T\in \mathbb{I}(i)$ is a feasible coalition containing $i$ and distinct from the coalition of all players $I$, i.e.
$$
F_i=\{\pi_i(z)\mid z\in F^T, i\in T\in \mathbb{I}, T\not=I\}.
$$
Here $\pi_i(z)$ denotes a natural projection of $z=(z_j)_{j\in T}\in F^T$ into $X_i$, where \mbox{$\pi_i(z)=z_i$}.
Thus, the set $F_i$ includes all strategies that person $i$ can apply where he joins any coalition distinct from the total one.

\begin{definition}
A coalition $S\in \mathbb{I}$~
{\bf \em $\alpha$-blocks} a strategic profile \mbox{$x\in F$} if $S$
has a joint feasible strategy $y^S\in F^S$ such that each member of $S$ strictly prefers the outcome (state) $(y^S,w^{-S})$ to the outcome $x$, regardless of the reaction $w^{-S}=(w_j)_{j\notin S}$ of the complementary coalition, where each $w_j$ belongs to the set $W_j=\conv F_j$.

The bundles $x\in F$ such that there is no coalition capable of $\alpha$-blocking $x$ forms the {\bf \em $\alpha$-core} of the game $G$.
\end{definition}
The above definition can now be reformulated in the following way:
If $S$ is a feasible coalition, then let $P^S$ be a
correspondence from $X$ to $X^S$, defined as
$$
P^S(x)=\{y^S \in X^S \mid {y^S} \times W^{-S} \subset \bigcap_{i\in S}\P_i(x)\},
$$
where ${y^S}\times W^{-S}= \{(y^{S}, w) \mid w \in W^{-S}\}$. Following this notation, a feasible
joint strategy $x\in F$ belongs to the $\alpha$-core if and only if there is no feasible coalition $S\in \mathbb{I}$ such that
$F^S\cap P^S(x)\not= \emptyset$.
Note that for every feasible coalition $S\in \mathbb{I}$ we have
$$
F^{-S} \subseteq W^{-S}\subseteq X^{-S}.
$$

Now we recall the notion of a ``balanced'' $n$-person game. It first appears in \cite{bon0,bon} and \cite{Shapley} later discussed it in the context of a game with transferable utility (a game with side-payments). Scarf  applied the notion of balancedness to a game in characteristic function form in \cite{Scarf0} and in \cite{Scarf} he applied it to a game in normal form.

Let $\Delta$ be the set of sets $\lambda = (\lambda_S)_{S\in \mathbb{I}}\in \R^\mathbb{I}_+$ satisfying the condition
$$
\forall i\in I,~~\sum_{S\in \mathbb{I},i\in S}\lambda_S=1~~\iff~~\sum_{S\in \mathbb{I}}\lambda_S\mathbf{1}_S=\mathbf{1}_I,
$$
where $\mathbf{1}_S(i)=1$ for $i\in S$ and $\mathbf{1}_S(i)=0$ otherwise.
The element $\lambda \in \Delta$ is called a balanced covering (or weights), and the family of coalitions $\{S \in \mathbb{I} \mid \lambda_S>0\}$ associated with it is called balanced.

\begin{definition} A generalized normal form game
$$
G=\la I,\{X_i,\P_i(\cdot)\}_{i\in I}, F, (F^S)_{S\in \mathbb{I}}\ra
$$
is called {\bf \em balanced} if for every balanced set $\lambda\in \Delta$, and every set of feasible strategies $y^S\in F^S$ with $S\in \mathbb{I}$ and $\lambda_S>0$, the state
$z = (z_i)_{i\in I}$ defined as
\begin{equation}
\forall i\in I~z_i=\sum_{S\in \mathbb{I}:i\in S}\lambda_Sy_i^S,
\end{equation}
is {\bf \em feasible}, i.e., $z\in F$.
\end{definition}

Finally, we will also assume a specific type of continuity of preferences.
\begin{definition} A generalized game $G$ is called $\alpha$-continuous if
for every feasible coalition $S\in\mathbb{I}$ the correspondence $P^S$ has open lower sections in $X$, that is, for every $i\in S$ and every feasible strategy $y^S\in F^S$,
the set $\{x \in X \mid {y^S}\times W^{-S}\subset \P_i(x)\}$ is open in $X$.
\end{definition}
Notice this is true if $\P_i(\cdot)$ has open graph for every $i\in I$. So, the assumption is reasonably unrestrictive.

\section{Theorem on Non-emptiness of $\alpha$-core}
\begin{theorem}
Under assumptions A1-A3 if the game
$$
G=\la I,\{X_i,\P_i(\cdot)\}_{i\in I}, F, (F^S)_{S\in \mathbb{I}}\ra
$$
is   $\alpha$-continuous and balanced, then $\alpha$-core is nonempty, i.e. $\mathcal{C}_\alpha(G)\not=\emptyset$.
\end{theorem}

\begin{proof} By contradiction. Assuming $\mathcal{C}_\alpha(G)=\emptyset$ we show that  does exist a continuous mapping $f:F\rta F$ that has no fixed point, which contradicts Brouwer's theorem. We describe the construction of this mapping below.

The assumption $\mathcal{C}_\alpha(G)=\emptyset$ means that for each $x=(x_i)_{i\in I}\in F$ there exists a coalition $S\in \mathbb{I}$ that blocks $x$, i.e.
$$
\exists y^S\in \P^S(x)\cap F^S\not=\emp.
$$
Moreover, by the assumption of $\alpha$-continuity of $G$, for some $V(x)\subset X$ open in $X$, one has
$$
x\in V(x)~~\&~~\forall x'\in V(x)~~y^S\in \P^S(x')\cap F^S.
$$
Here $V(x)$ is an open neighborhood of $x\in F$. Since $V(x)$ form a covering of the compact set $F$, there is a finite subcovering $V_k=V(x^k)$, and the corresponding coalition $S_k\subseteq I$, $\alpha$-blocking every $x\in V_k$, $k=1,2,\ldots,m$. Next, we consider the continuous decomposition of the identity corresponding to the family $V_k$, i.e., we find a set of continuous functions $\eta_k:F\rta [0,1]$ such that
$$
\forall x\in F~\sum_{k=1}^m \eta_k(x)=1~~\&~~\eta_k(x)>0\iff x\in V_k,~~k=1,2,\ldots,m.
$$
Now, for an arbitrary chosen and fixed $z=(z^{S_k},z^{-S_{k}})\in \prod_{i\in I}F^{\{i\}}$, we define on $F$ the function $h(\cdot)$ by the formula
$$
h(x)=\sum_{k=1}^m \eta_k(x)(y^{S_k},z^{-S_{k}}).
$$
The continuity of $h(\cdot)$ follows from the construction. Next, we specify
$$
\theta(x)=\min_{i\in I}\sum_{k: i\notin S_k} \eta_k(x)<1
$$
and consider
$$
f(x)=\frac{h(x)-\theta(x)z}{1-\theta(x)}.
$$
The function $f(x)=(f_1(x),\ldots,f_n(x))$ is the desired one; we will show that it has the required properties. One has
$$
\forall i\in I ~~~f_i(x)=\sum_{k: i\in S_k} \lambda_k(x) y_i^{S_k}+\lambda_{m+i}(x)z_i,
$$
where
$$
 \lambda_k(x)=\frac{\eta_k(x)}{1-\theta(x)}~~~\&~~~
\lambda_{m+i}(x)=\frac{\sum_{k: i\notin S_k} \eta_k(x)-\theta (x)}{{1-\theta(x)}}.
 $$
By construction, the set $\lambda=\{\lambda_\xi\}_{\xi=1}^{m+n}$ forms a balanced set of weights for the family of coalition strategies $y^{S_k}\in F^{S_k}$, $k=1,\ldots,m$ and $z_i\in F^{\{i\}}$, $i=1,\ldots,n$.
The assumption of a balanced game implies $f(x) \in F$.

Moreover, there exists an individual $i\in I$ such that
$\theta(x)=\sum_{k: i\notin S_k}\eta_k(x)$ $\Ra$ $\lambda_{m+i}(x)=0$. Therefore, we can write the components of $f(x)$
as
$$
f_i(x)=\sum_{k: i\in S_k} \lambda_k(x) y_i^{S_k},
$$
and for other $j\in I$, $j\not=i$ we have
$$
f_j(x)=\sum_{k:i,j\in S_k}\lambda_k(x) y_j^{S_k}+\sum_{k:i\in S_k,j\notin S_k}\lambda_k(x)w_j(x),
$$
where
$$
w_j=\frac{\sum_{k: j\in S_k,i\notin S_k}\lambda_k(x)y_j^{S_k}+\lambda_{m+j}(x)z_j}{\sum_{k: j\in S_k,i\notin S_k}\lambda_k(x)+\lambda_{m+j}(x)}\in F_j.
$$
Since $\sum_{k: j\in S_k,i\notin S_k}\lambda_k(x)+\lambda_{m+j}(x)=\sum_{k:j\notin S_k,i\in S_k}\lambda_k(x)$,
then $f(x)$ is represented as a convex combination of vectors with numbers $k:i\in S_k$ such that for $w^{-S_k}=(w_j)_{j\notin S_k}\in W^{-S_k}$ we have
$$
  (y^{S_k}, w^{-S_k})\in \bigcap_{j\in S_k}\P_j(x)\subseteq \P_i(x).
$$
Therefore, $f(x)\in \conv \P_i(x)$. However, now if we assume that there is a fixed point $\bar x=f(\bar x)$, then for this player,
we get a contradiction with the assumed in A3 convex irreflexivity: $\bar x\notin \conv\P_i(\bar x)$, which is what we had to prove.
\end{proof}

The following example illustrates the principle underlying the construction of the function $f(x)$ that yields the desired contradiction.


\begin{example}\rm
Consider a three-player game, and let \mbox{$x=(x_1,x_2,x_3)\in F$} be an out\-come that is blocked by coalitions $S_1=\{1,2\}$, $S_2=\{1,3\}$, and $S_3=\{2,3\}$. Suppose $\eta_1(x)=1/2$, $\eta_2(x)=1/3$ and $\eta_3(x)=1/6$ are the coefficients of the unity de\-com\-po\-si\-tion, defining
$$
h(x)=\frac{1}{2}(y_1^{S_1},y_2^{S_1},z_3)+\frac{1}{3}(y_1^{S_2},z_2,y_3^{S_2})+
\frac{1}{6}(z_1,y_2^{S_3},y_3^{S_3})=
$$
$$
\left ( \frac{1}{2}y_1^{S_1}+\frac{1}{3}y_1^{S_2}+\frac{1}{6}z_1,
\frac{1}{2}y_2^{S_1}+\frac{1}{6}y_2^{S_3}+\frac{1}{3}z_2,
\frac{1}{3}y_3^{S_2}+\frac{1}{6}y_3^{S_3}+\frac{1}{2}z_3\right).
$$
Here $z=(z_1,z_2,z_3)\in F^{\{1\}}\times F^{\{2\}}\times F^{\{3\}}$ is an arbitrarily chosen element.
Next we find $\theta(x)=1/6$ and specify
$$
f(x)=\frac{h(x)-1/6z}{1/2+1/3}=
$$
$$=
\left ( \frac{3}{5}y_1^{S_1}+\frac{2}{5}y_1^{S_2},
\frac{3}{5}y_2^{S_1}+\frac{1}{5}y_2^{S_3}+\frac{1}{5}z_2,
\frac{2}{5}y_3^{S_2}+\frac{1}{5}y_3^{S_3}+\frac{2}{5}z_3\right)=
$$
$$
 \frac{3}{5}\left (y_1^{S_1},y_2^{S_1}, \frac{1}{3}y_3^{S_3}+\frac{2}{3}z_3\right)
 +\frac{2}{5}\left (y_1^{S_2}, \frac{1}{2}y_2^{S_3}+ \frac{1}{2}z_2,y_3^{S_2}\right).
$$
Received:
$$
\left (y_1^{S_1},y_2^{S_1}, \frac{1}{3}y_3^{S_3}+\frac{2}{3}z_3\right)\in \P_1(x)\cap\P_2(x)~~\&
$$
$$%~~
\left (y_1^{S_2}, \frac{1}{2}y_2^{S_3}+ \frac{1}{2}z_2,y_3^{S_2}\right)\in \P_1(x)\cap \P_3(x),
$$
whence $f(x)\in \conv \P_1(x)$. Moreover, $f_1(x)=\frac{3}{5}y_1^{S_1}+\frac{2}{5}y_1^{S_2}$,
$f_2(x)=\frac{3}{5}y_2^{S_1}+\frac{1}{5}y_2^{S_3}+\frac{1}{5}z_2$ and $f_3(x)=\frac{2}{5}y_3^{S_2}+\frac{1}{5}y_3^{S_3}+\frac{2}{5}z_3$. It follows that the balanced covering implementing $f(x)$ is the family $\{S_1,S_2,S_3,\{2\},\{3\}\}$, implemented by the set of weights: $\lambda_{12}=3/5$, $\lambda_{13}=2/5$, $\lambda_{23}=1/5$, $\lambda_{2}=1/5$, $\lambda_{3}=2/5$.
\end{example}

\subsection{Conclusion} The paper presents a simple short proof of an important game-theoretical concept of $\alpha$-core under general  assumptions, known in literature. We prove
only a finite-dimensional theorem as soon as only at this part our achievements have been made.  The general theorem---the case of Hausdorff linear space as a space of strategies can be proven repeating arguments of \cite{M-Y}. Moreover, one can hope that the  method  similar to applied here can be developed to effectively study existence of other core concepts; e.g. it may be $\beta$ or $\gamma$ cores, that still have not known good conditions under which they exist.


\bigskip

\end{document}
